

Response to reviewers:

Response to reviewers 2:

1. We confirmed the diagnosis of NASH in all patients. Therefore, we used NASH instead of NAFLD in manuscript.

2. We mentioned about statistically insignificant in 'conclusion section (the last paragraph of discussion)

Response to reviewer 3: We mentioned in both introduction and in discussion section about recently published literature related to bone mineral density and NASH. We also changed some part of discussion section (showed in red color).